Author's response to reviews

Title: Micro-geographical variation in exposure to Schistosoma mansoni and malaria, and exacerbation of splenomegaly in Kenyan school-aged children

Authors:

Dr Mark Booth (mb350@cam.ac.uk)
Birgitte J Vennervald (bjv@bilharziasis.dk)
LeeCarol Kenty (lck24@cam.ac.uk)
Anthony E Butterworth (butterworth@vet.uz.ac.zw)
Henry C Kariuki (schisto@wananchi.com)
Hilda Kadzo (jmwatha@nairobi.mimcom.net)
Edmund Ireri (jmwatha@nairobi.mimcom.net)
Clifford Amaganga (jmwatha@nairobi.mimcom.net)
Gachuhi Kimani (gkimani@nairobi.mimcom.net)
Joseph Mwatha (jmwatha@nairobi.mimcom.net)
Amos Otedo (jmwatha@nairobi.mimcom.net)
John H Ouma (ouma@wananchi.com)
Eric Muchiri (schisto@wananchi.com)
Dr David W Dunne (dd@mole.bio.cam.ac.uk)

Version: 4 Date: 20 Apr 2004

Dear BMC

We are hereby submitting the final draft of our paper entitled 'Micro-geographical variation in exposure to Schistosoma mansoni and malaria, and exacerbation of splenomegaly in Kenyan school-aged children'. Typographical errors have been corrected, and the following minor amendments to the manuscript have been made.

. The last paragraph in the discussion has been separated into a section entitled 'conclusions'

. IgG3 'titres' have been replaced with 'levels' throughout.

. The correct statement in the competing interests section has been made

We look forward to seeing the paper published.

Yours sincerely

Mark Booth